Author's response to reviews

Title: Fetal demise and associated factors following umbilical cord prolapse in Mulago hospital, Uganda. A retrospective study.

Authors:

Esau Wangi Wasswa (wweb2@yahoo.com)
Sarah Nakubulwa (sarahug@gmail.com)
Twaha Mutyaba (tmutyaba@yahoo.com)

Version: 4 Date: 30 January 2014

Author's response to reviews:

Deleted from the abstract and introduction, the sentence "There is no literature from Uganda regarding this pregnancy complication"

Deleted the sentence "It is surprising that a simple intervention like inflation of the bladder while awaiting surgery was not being used in this setting"